Contents lists available at ScienceDirect



Renewable and Sustainable Energy Reviews

journal homepage: www.elsevier.com/locate/rser



Overview of research on China's transition to low-carbon development: The role of cities, technologies, industries and the energy system



Chen Wang^{a,b,c,*}, Anita Engels^d, Zhaohua Wang^{b,c}

^a Donglinks School of Economics and Management, University of Science & Technology Beijing, 100083, China

^b School of Management and Economics, Beijing Institute of Technology, Beijing 100081, China

^c Center for Energy & Environmental Policy Research, Beijing Institute of Technology, Beijing 100081, China

^d Centre for Globalisation and Governance, Universität Hamburg, Germany

ARTICLE INFO

Keywords: Low-carbon cities Low-carbon technologies and industries Low-carbon transition of China's energy system

ABSTRACT

China is experiencing a transition to low-carbon economic development. This paper assesses the current literature on the potentials of and barriers to China's transition to low-carbon development, identifying promising fields of action and suggesting a research agenda that systematically addresses the shortcomings. Through a broad literature review, we select three main research areas of interest: low-carbon cities, low-carbon technologies and industries, and the transition of China's energy system. As an innovative work, we also summarize some specific issues discussed more in Chinese journals but less in English language ones. Some elements of a more comprehensive research agenda that can improve the understanding of China's ability to enter a low-carbon development pathway are suggested.

1. Introduction

China's crucial role as an emitter of greenhouse gases has long been recognized in the debate about global climate change mitigation. Researchers have emphasized both the need for a significant reduction in China's emissions and the potential for accomplishing this if the Chinese government agrees to a fundamental policy shift. More recently, it has become increasingly acknowledged that this policy shift has already started to occur, even though it may be a side effect of China's domestic energy issues rather than a direct result of a global climate agreement. However, the economic, political and institutional preconditions that would enable China to begin a transition to a low-carbon development pathway are still poorly understood. Whereas some researchers have overestimated the steering and implementation capacities of a central government in a planned economy [142], others have underestimated the societal dynamics that might promote such a transition at a rate faster than that in other countries. This paper therefore assesses the current literature on the potentials of and barriers to China's transition to low-carbon development. Through a broad literature review, we selected some of the best covered topics. Although we omitted a number of topics, we believe that we have covered those most relevant to the overall question of low-carbon development in China. The objective of this review paper is to combine existing literature from various disciplines to systematize current knowledge about the most important challenges of low-carbon development in China and to discuss the implications of the findings for finding promising routes to strengthen low-carbon strategies and develop research on this topic further. Obviously the literature covers a wide range of - sometimes incompatible methodologies. For the sake of transparency, we mention these methodologies, but it is beyond the scope of this review paper to discuss in depth their respective underlying assumptions and epistemological starting points. As this is a review paper in nature, it cannot directly contribute to the reduction of GHG emissions, but we hope to improve the understanding of how such strategies are - or are not - leading to emission reductions in the long run. The paper is organized as follows: in the remainder of this introduction, we explain the preconditions for low-carbon development in China and deduce the three strands of literature on lowcarbon development in China that have been most intensely researched in the past years. Chapters 2-4 follow the three themes we have identified: Chapter 2 summarizes the literature on low-carbon cities in China, Chapter 3 focuses on the literature about low-carbon technologies and industries, and Chapter 4 reviews the literature on the transition of China's energy system. While these three chapters focus on research publications in Anglophone mainstream journals, we add a brief fifth chapter on the literature published in Chinese journals. Chapter 6 summarizes our findings and draws some conclusions.

What are the preconditions for low-carbon development in China? Even though China's per capita CO_2 emissions remain below those of other industrialized countries, at 6.7 t (2010–2014), its total CO_2

http://dx.doi.org/10.1016/j.rser.2017.05.099 Received 10 May 2016; Received in revised form 26 February 2017; Accepted 18 May 2017 Available online 08 July 2017 1364-0321/ © 2017 Elsevier Ltd. All rights reserved.

^{*} Corresponding author at: School of Management and Economics, Beijing Institute of Technology, Beijing 100081, China. *E-mail address:* cherylwang@bit.edu.cn (C. Wang).

emissions have grown enormously in recent decades, thus making China the world's largest single national emitter [108]. China's energy system is mostly based on coal [91], and the majority of its coal-fired power plants operate at relatively low efficiency levels. Energy security has become a pressing issue in terms of both domestic and international strategies [42,43]. Domestic problems connected to energy consumption are becoming increasingly important. At the same time, there is pressure from rural areas because China still suffers from rural energy poverty and from large metropolitan areas, where the population suffers from poor and sometimes dangerous urban air quality. As a reaction to all these pressures and in an attempt to become a more active player in international climate negotiations. China's government has recently made a voluntary commitment to lowering its CO₂ emissions per unit of GDP by 60-65% by 2020 (compared with 2005 levels) and to increasing the share of non-fossil fuels in its primary energy mix (UNFCCC 2015; [81]). Another recently established goal is that the share of non-fossil fuels used in primary energy consumption should be increased to 15% by 2020 [82]. China has also issued a series of national strategies for energy conservation and emission reductions; these are presented in the 12th Five-Year Plan.¹ Policies and measures such as emissions trading have been introduced in several provinces [18,39]. The societal preconditions required for implementing such policies effectively and for achieving ambitious mitigation targets are therefore a pertinent research topic.

This article seeks to contribute to the above-mentioned research area and provides a literature review, on the basis of which research gaps will be identified. Our main focus is on mainstream peer-reviewed journals in English, but we also examine relevant debates in peer-reviewed journals in Chinese. A review of research papers in relevant journals² revealed three main research areas of interest: low-carbon cities, low-carbon technologies and industries, and the transition of the national energy system from one based on fossil fuels toward one based on renewable and nuclear energy sources. The three research areas resonate with three important economic and social dynamics currently affecting China: rapid urbanization, rapid technological development and industrialization, and an ongoing increase in energy demand. We assess these three strands of literature consecutively, summarizing recent findings and identifying research gaps. The final chapter therefore identifies what appears from the literature as promising approaches to low-carbon development in the Chinese context and suggests elements of a more comprehensive research agenda that can improve the understanding of China's ability to enter a low-carbon development pathway.

2. Current research on low-carbon cities in China

As densely populated and resource-intensive regions, cities have always been hotspots of innovation. Since the beginning of industrialization, however, cities have been under intense environmental pressure because of their high concentrations of industry, infrastructure and population. They have also been regarded as the world's main sources of greenhouse gas emissions and the primary contributors to global warming and climate change [14]. According to UN Habitat [96], approximately 40-78% of global GHG emissions are estimated to originate in cities. Moreover, the proportion of greenhouse gas emissions from cities is increasing continuously because of ongoing urbanization [65]. In many parts of the world, urban sprawl and energy-intensive development patterns are still seen as pathways to urbanization [1.75]. This phenomenon is particularly true for some developing countries, whose energy consumption and GHG emissions are expected to continue to increase significantly along with rising living standards and material affluence [22,78].

As the world's largest developing country, China is experiencing urbanization on an unprecedented scale. China's urbanization rate has increased from approximately 36% in 2000 to nearly 53% in 2012 [74]. Migration, urban expansion, and the emergence of new cities near existing cities all indicate that urbanization in China will exert continuing acute pressure on infrastructure, economic growth, land development, urban resource demands, and pollution [9]. China's 35 largest cities contain approximately 18% of the country's population and contribute 40% of the country's energy usage and CO₂ emissions [14,44]. This percentage continues to grow along with China's highspeed urbanization.³ A low-carbon urban development strategy would therefore offer a high GHG emission-reduction potential for China. At the same time, China's cities are extremely vulnerable to the impacts of climate change because of their high population densities, concentrated economic activities and scarce natural resources. Cities in northern and western China tend to experience droughts, dust storms, and smog, which have been particularly serious in recent years, whereas the eastern and southern cities often suffer from floods and extreme rainstorms. Liu and Deng have shown that the mean annual precipitation is expected to decrease by 2-10% in drought-prone northern China but increase by more than 20% in flood-prone southern China [65]. A strong dust storm attacked the northwest provinces in April of 2014, including some important cities such as Lanzhou City in Gansu Province and Wulumuqi City in Xinjiang Province. Some eastern cities, such as Shanghai, and southern cities, such as Guangzhou, suffer from rainstorms almost every summer. It can be assumed that the immense scale of urbanization will increase public pressure and therefore lead to a bolder and more aggressive decarbonization strategy for both national and regional policy makers.

Which low-carbon city initiatives have been started in China, and what can we learn about them from the existing literature? An initial low-carbon city program was implemented by China's Ministry of Construction and the Worldwide Fund for Nature in 2008. In August 2010, China's National Development and Reform Commission (NDRC) launched a low-carbon city experimental project that was implemented in eight cities: Tianjin, Chongqing, Shenzhen, Xiamen, Hangzhou, Nanchang, Guiyang, and Baoding⁴ [16]. The concept of the low-carbon city, which integrates elements of both a low-carbon economy and a low-carbon society, has been found to be a new path by which China can achieve its goals of sustainable urbanization, ecological civilization and scientific development [106,14]. Therefore, these "low-carbon cities" are intended to develop low-carbon economies and to promote low-carbon lifestyles. Specifically, a low-carbon city is characterized by reduced pollution, low emissions, and high energy efficiency [65].

So far, however, low-carbon cities in other countries have been little more than an inspiring guiding principle for city administrators and citizens. Even if policies promoting low-carbon cities abound, it would

¹ The Five-Year Plans, as an important part of Chinese domestic economic plans, are a series of social and economic development initiatives that mainly focus on carrying out major national construction projects, mapping strategies for economic development, setting growth targets and launching reforms. The first Five-Year Plan began in 1953. The 11th Five-Year Plan covers the period 2006-2010, and the 12th covers 2011-2015.

² Using the SCI and SSCI network databases, we searched with the key words "lowcarbon" and "China" to identify related articles published during the 2004-2016 period. The total number of articles identified was 16926. Because we are focusing primarily on the effects of low-carbon policies on China's economy and society from the perspectives of management and economics, papers focusing on technologies from purely natural science or technological perspectives were excluded from our list. For similar reasons, papers about low-carbon agriculture were also beyond the scope of our review because that field is another vast and complex research area closely related to chemistry, biology, geology, etc. Moreover, by using Bibexcel software, we also analyzed the frequency of key words appearing in the 6000+ articles we identified. In addition to obvious words such as "China", "carbon dioxide", "low-carbon", and "climate change", key words closely related to content included issues such as "low-carbon economy", "energy consumption", "renewable energy", "energy efficiency", and "low-carbon city". Therefore, our paper is divided into reviews centered on the concepts of "low-carbon city", "low-carbon technologies and industries in China", and "the transition of China's energy system".

³ By 2014, China's urban population had grown to 54% of the country's total

population (World Bank: http://data.worldbank.org/indicator/SP.URB.TOTL.IN.ZS). ⁴ Source: http://www.worldbank.org/en/news/feature/2012/05/03/sustainable-lowcarbon-city-development-in-china.

Table 1

Definition of scope	1, 1	2, an	d 3	GHG	emissions
(Source: [41,2])					

	Definition of scope
Scope 1: Direct GHG emissions Scope 2: Indirect GHG emissions from electricity	Direct GHG emissions occur from stationary combustion, mobile combustion, process emissions and fugitive emissions. These are emissions from the generation of purchased electricity consumption, steam and heating consumption.
Scope 3: Other indirect emissions	Emissions in this optional reporting capacity are a consequence of other life-cycle emissions excluded from scopes 1 and 2 (for example, GHG emissions generated during the process of raw materials production within district boundaries of the city)

be overly optimistic to expect that local governments have sufficient legal and political leeway and steering capacity to achieve such ambitious goals easily [3,4]. Financial limitations are also common, e.g., the lack of resources to invest in low-carbon technologies [71].

What can be learned from the available research on the current status of low-carbon cities in China? Current research mainly focuses on the following topics: first, the challenges of establishing GHG emissions inventories; second, the identification of factors influencing GHG emissions in China's cities; and, third, the exploration of paths by which China's cities can achieve low-carbon development.

2.1. Establishing a GHG emissions inventory of China's cities

Some authors have suggested that developing benchmarks and a more comprehensive GHG emissions inventory is a necessary first step in the context of global efforts to mitigate climate change [24]. Establishing a local carbon emissions accounting system is a prerequisite to and the first priority in realizing a low-carbon transition [71]. The research on establishing a GHG emissions inventory has important theoretical and practical implications for China's efforts in energy conservation, emissions reduction, and low-carbon city construction. On the one hand, transparent and robust quantification is the basis against which the effects of mitigation efforts can be measured. GHG inventories should therefore be conducted regularly to track progress and indicate areas for further improvement [90]. On the other hand, a GHG emissions inventory that is in line with up-to-date international GHG inventory standards can help to facilitate comparison and analysis with other cities in the world, which is important in providing central and local policy makers with a clear understanding of their city's GHG emissions status.

A comprehensive GHG emissions inventory requires two important components: a standard inventory methodology and up-to-date activity data. Kennedy et al. [41] have elaborated on the methodology of GHG emissions inventories in detail. In general, the methodology can be divided into two categories: the top-down approach and the demandcentered, bottom-up approach. Moreover, three scopes have been suggested for GHG inventories at the city level (Table 1).

Although there are several studies of GHG emissions in Chinese cities, so far, most have been based on the city's aggregate energy consumption by using a top-down methodology [2], in line with IPCC guidelines and the WRI/WBCSD model.⁵ Some authors have discussed conceptual problems and problems of data availability that arise when applying this methodology. Dhakal [14] has described a further methodological problem that has to do with differences between official *city* populations and *urban* populations: cities and towns are politico-administrative units. Designating a place as a "City" has historical, political, resource distribution, and other implications in China. Generally, the statistical data of a certain city cover both urban and

rural areas in that city. The best method for estimating China's urban energy consumption would be to determine the energy consumption of each city and town and then divide these figures into urban and rural energy uses. However, the energy use information that would be required for this approach is available for only a few large cities. Therefore, so far, most studies on urban energy use in China rely on an aggregated and top-down methodology. In this sense, Sugar et al. [90] have provided comprehensive and detailed emissions inventories for Shanghai, Beijing, and Tianjin and have compared the results with those for ten other cities around the globe. The results show that the majority of emissions in the three Chinese cities were from electricity production, heating and industrial fuel use, and ground transportation. When compared with other cities around the world, these three cities are among the highest per capita emitters. The authors have also examined similarities and differences in GHG emissions among cities in China and cities located in other countries. The results provide a benchmark for discussions of the effectiveness of strategies designed to reduce carbon emissions.

Inventories using a bottom-up methodology are still rare. The study of Bi et al. [2] is the first comprehensive accounting system for GHG emission inventories at the city scale in China; it is based on a bottom-up methodology and includes six sectors. This study focuses on scope 1 and 2 emissions (see Table 1) in Nanjing City. The results show that the three largest GHG contributors are industrial energy consumption, industrial processes, and transportation. In contrast to a top-down methodology, the bottom-up approach results in lower per capita and per GDP carbon emissions values (in this particular study, such values were comparable to or even lower than the world average). Wang et al. [106] have suggested that although the GHG inventories calculated by a topdown approach can provide proxies of total GHG emissions, they do not provide sufficient information for local governments to define operable measures to reduce carbon emissions. Analyzing the GHG emissions characteristics of China's mega-cities based on the study of Bi et al. [2], Wang et al. have developed inventories for 12 of the most developed cities in China by using bottom-up approaches (similarly to Bi et al. [2]). The results show that although average per capita emissions in these cities are lower than those in many large cities in developed countries (London, Los Angeles, New York, etc.), the total carbon emissions of Chinese cities can still be well above those of their Western counterparts. Tong et al. [95] have quantified GHG emissions associated with four different-sized cities in China by a method called community-wide infrastructure GHG footprint. This method can represent the life cycle and transboundary impacts of infrastructure use. The research has analyzed the GHG emissions in Yixing, Qinhuangdao, Xiamen and Beijing by calculating the carbon footprints of seven of their most essential infrastructures, i.e., electricity, non-electricity energy, water supply and wastewater treatment, transportation, municipal waste management, construction materials, and food. The results show that industrial energy use is the dominant contributor to GHG emissions in the four cities [95].

GHG emissions inventories in China's cities often cover scopes 1 and 2, whereas scope 3 emissions are rarely included (see Table 1). For example, Li et al. [50] have outlined a detailed inventory of CO_2

⁵ The 2006 IPCC Guidelines for National Greenhouse Gas Inventories provides detailed instructions on how to model this. See http://www.ipcc-nggip.iges.or.jp/public/2006gl/index.html. The WRI/WBCSI model is the Greenhouse Gas Protocol (GHGP) developed by the World Resources Institute and World Business Council on Sustainable Development (WBCSD). See http://www.wri.org/our-work/project/greenhouse-gas-protocol/online-courses#project-tabs.

emissions and future trends as a basis on which Shanghai might determine actions to help mitigate climate change. This paper estimates the inventory of CO2 emissions from energy uses (including industrial activities) but excludes other indirect emissions. A study by Song et al. [88] is among the few studies that have extended beyond this limited scope, although the research focused on only the community level (the smallest official social unit in a particular area or place, often consisting of only 4000-15,000 inhabitants). Through life-cycle analysis, the authors have presented a three-scope accounting framework for community carbon emissions, including emissions from direct fossil fuel combustion, purchased energy (electricity, heat, and water), and supply chain emissions embodied in the consumption of goods. A typical high-quality community in Beijing is used as a case study to elaborate on the CO2 emission inventory in detail. The results show that emissions from different elements in the supply chain embodied in the consumption of goods are too significant to be ignored. The top three emission items are transport and communication (41.36%), buildings (14.11%), and education and recreation (10.41%). The emissions stemming from buildings are a conservative estimation because this study considered only major material consumption owing to a lack of data. Moreover, this study shows that energy consumption tends to increase along with increases in income [88].

In summary, the GHG inventories conducted in China to date remain at the initial stages and are very limited in scope:

First, because of differences in the concepts, scopes, data availabilities and statistical methods between Chinese and Western cities, it is difficult for China's cities to directly use the common international methodology to establish GHG inventories. China would have to establish GHG inventories in line with international standards to improve comparability.

Second, the GHG inventories carried out to date lack continuity. Most of the research has focused on one or only a few cities in a given year. Few of these studies have established a basis for long-term monitoring of GHG emissions, which is important in analyzing changes.

Third, China's GHG inventories seldom cover scope 3 emissions [88]. This is partly because life-cycle analysis is often used to add scope 3 emissions, but it is difficult to obtain related data in China. Publication of more emissions data will require long-term adjustments by the government, enterprises, and other organizations.

Overall, there seems to be a research gap concerning the use of inventories in the policy process and in other aspects of governance, although Sugar et al. [90] have at least mentioned governance as a topic. Some inventories include policy implications, but their actual role in the policy process is poorly understood. Here, one can draw insights from a series of recent studies on the challenges of local governance, e.g., the study by de Jong et al. [13], which explains the substantial implementation gap by a top-down technocratic planning mode that fails to acknowledge the close interrelations between local government actors and land and real estate developers. A comparison of Chinese and Swedish eco-cities emphasizes the need to overcome typical cross-sector barriers for local governance attempts to achieve environmental goals [117]. Recent debates on urban governance in Chinese megacities have also revealed a shift in governance from clientelistic to more corporatist modes. Although this shift has also included more transparent policy making and a stronger role for the district government, the central government often interferes in unforeseeable ways [145,146]. The complex interactions between the central government and the local government remain an important precondition for developing low-carbon strategies at the city level. Although the central government issues far-reaching decarbonization targets, it often remains unclear how city governments can actually contribute to achieving these goals and how local development goals can be reconciled with low-carbon development pathways.

2.2. Identification of factors influencing GHG emissions in China's cities

Developing an understanding of the factors influencing GHG emissions in cities is a crucial prerequisite for mitigating China's anthropogenic emissions, as policies can then better prioritize and identify the most important leverage points. Quantifying the magnitude of each factor that drives emission changes in cities is necessary so that a potential bottom-up climate mitigation policy at the city and sectoral levels can be designed and initiated. Li et al. [54] have used Wuhan City as an example and have studied the driving factors of GHG emissions. In this paper, they reveal the dynamic relationships between the city's carbon footprint and the factors influencing it. The results show that population growth and economic development are the main influencing factors for GHGs in Wuhan City. Wang et al. [106] have empirically studied the influences of urbanization processes, economic factors, industry structure, and energy intensity on GHG emissions. Notably, they have also considered the share of tertiary industry and R&D output as influencing factors. The results show that both influence GHG emissions negatively, leading the authors to conclude that the city administration of Beijing should pay more attention to the typically underestimated tertiary sector. Li et al. [44] have selected Tianjin in northern China as a case study to identify the factors driving CO₂ emissions during the 1996-2012 period. In addition to basic factors such as energy intensity, population, and income, they have also considered the effect of foreign direct investment (FDI). The results show that the influx of foreign capital harms the local environment. As an economic center and as the largest coastal city in northern China, Tianjin is an important investment hotspot for foreign capital. FDI has had an important effect on economic and technological development in Tianjin City, but this goes hand in hand with accelerated deterioration of the local environment. Investors see relocating their polluting industries to Tianjin City as a way of avoiding more stringent environmental regulations elsewhere [44]. However, this result is inconclusive, as Wang et al. found no significant influence of foreign direct investment on China's carbon intensity.

Many scholars have studied the factors influencing cities' GHGs from a purely economic perspective. The factors usually include carbon/energy intensity, power mixture, industrial structure, and population. Wang et al. [99] have analyzed the main drivers of increasing GHGs in Suzhou City during the 2005–2010 period. Chong et al. [9] have investigated the key drivers affecting emission increases in six Chinese megacities: Beijing, Shanghai, Tianjin, Chongqing, Guangzhou, and Hong Kong. In all cities, economic and energy intensity effects have long been the two dominant factors contributing to changes in carbon emissions.

Some scholars have also studied this issue from a broader perspective by taking urban development intensity into consideration. Urban development intensity can be defined as comprising the multiple impacts that human development activities exert on urban areas, including land use, population, economic intensity, the spatial distribution of infrastructure, and public service facilities. Wang et al. [102] have gualified the relationship between urban development intensity and carbon dioxide emissions. They have suggested that improved urban planning and spatial optimization may be effective means of reducing GHG emissions in China's cities because factors such as land-use intensity, economic intensity, population intensity, infrastructure intensity, and public service intensity are all drivers of GHG emissions. Land use intensity is an index for measuring the spatial pattern of land use. It is often reflected by indicators such as the ratio of built district to urban areas, the ratio of living land use to built district areas, the ratio of production land use to built district areas, and the ratio of infrastructure land use to built district areas. Higher land use intensity refers to a larger proportion of built-up areas. With relatively higher populations and levels of construction, large built-up areas have exerted more pressure on both the environment and the

economy, resulting in further emissions growth. This study concludes that, when pursuing low-carbon strategies, policy makers should consider the effects of urban development intensity, for example, by controlling land-use intensity. It might be debatable that tighter population and construction will cause environmental pressure because higher intensity also leads to less spatial demand; moreover, densely populated cities are often thought to be environmentally preferable, at least compared with suburban sprawl developments, because of their higher intensity. However, in China's cities, there are certainly some cases of inefficient land use. One example would be the so-called "ghost cities", which consist of residential high-rise buildings without anyone living in them. Even though these cities do not have high energy demand because of the lack of population, the emissions stemming from the construction process would still have to be taken into account. Wang et al. [102] also mention that China's continued family planning policy contributes to reducing population intensity; this also controls the scale of urban populations and helps to avoid increases in CO2 emissions. Without delving into ethical considerations of forced birth control, it should be mentioned that this policy has recently been changed for demographic reasons to allow two children per family. Zhao et al. [130] have estimated the carbon emissions of different land use types in Nanjing City. By establishing matching relationships between different land use types and GHGs, the carbon emission intensity of regional land use in this city was analyzed. The results suggest that to decrease the carbon emission intensity of land use, it is important to consider spatial expansion, intensity, and structure in future land use and urban development. Carbon emission effects should also be introduced into land use planning.

Among all studies of the factors influencing GHGs in cities, the STIRPAT model is a particularly popular method. STIRPAT is an acronym for Stochastic Impacts by Regression on Population, Affluence, and Technology. It refers to a statistical and conceptual model for assessing human impacts on the environment. It has been adopted as an analytic strategy to identify the primary drivers of environmental harm and to uncover leverage points for ameliorating that harm. For example, Li et al. have used this model to study the driving factors of GHGs in Wuhan City [54]. From this model, Wang et al. have developed an improved STIRPAT model to explore the influence of the level of urbanization, level of economic development, proportion of industry, and energy intensity on GHG emissions [106]. Also using the STIRPAT model, Li et al. [44] have used Tianjin City as an example to identify factors influencing GHGs. Many other researchers have used the index decomposition analysis method to study this issue. This method is an analytical tool originating from energy studies and has been extended to areas other than energy and emissions. For example, Wang et al. [99] have decomposed GHGs into impact factors, including population, per capita GDP, carbon emission per unit of energy consumption, carbon intensity, and energy intensity, to study the roles of these factors in GHG changes. This method has also been used in a study by Chong et al. [9].

Other methodologies have also been applied to analyzing the drivers of GHG emissions. In managing the dynamical evolution and the intrinsic stochastic behavior of vital elements inside urban energy systems, Feng et al. [20] have used system dynamics (SD) to describe the inner interactions and structures affecting urban developments and to identify both desirable and undesirable interventions.

In summary, for the methodologies used to explore the factors influencing GHGs, the STIRPAT model and the index decomposition method are often applied. The distinctions among studies lie in the differences of sample cities, the uniqueness of factors, and data accuracy. The factors influencing GHGs can be summarized into four categories: the economic level, industry zoning, energy system development, and city layout. First, the GHG emissions are considered to increase along with economic growth, which makes strong policies to control emissions at the city level less attractive for local governments. However, the exact relation between economic level and GHG emission can be further studied by decoupling research, which has already attracted some researchers' attention. In a few of China's cities, especially the large ones, strong decoupling effects between economic development and GHG emissions have already appeared, but because of strong economic imbalances among China's cities, this cannot easily be generalized. Second, industry zoning has a few more factors such as industrial structure, the proportion of tertiary industry, investment, etc. Researchers often advocate the adjustment of industrial structure. However, for many cities, the rapid increase of GHG emissions is caused by the development of the tertiary sector instead of the industrial sector. Researching emissions from the tertiary and household sector perspective may be a new approach to controlling GHG emissions in cities. Third, the energy system and energy intensity are crucial to emissions and are discussed in Section 4 of our paper. For the last point, the city layout is very important, but there is still a lack of detailed academic discussion, and improved city designs are particularly difficult to achieve in already existing cities.

2.3. Exploring paths to building low-carbon cities

The establishment of low-carbon cities has been promoted worldwide because cities are key drivers of energy usage and associated carbon emissions. The concept of the low-carbon city, which integrates elements of both a low-carbon economy and a low-carbon society, provides a new model of sustainable urbanization for China, a model directed toward ecological civilization⁶ and scientific development [127]. However, even if governments develop a basic understanding about the most important features of a low-carbon city, the most difficult question is the transition: how to switch from a traditional development path to a low-carbon development path.

Zhang et al. [127] have presented a scenario analysis of Beijing's future energy consumption and GHG emissions. They have used the Long-range Energy Alternatives Planning (LEAP) model to simulate a range of pathways and analyze how these would change energy consumption and carbon emissions from 2007 to 2030. They have suggested that although regulation and technological improvements will play an important role in mitigating GHG emissions in Beijing, these factors would not be sufficient to cause larger reductions. Larger reductions would require a transformation of the city's economic and social development pattern, including clean energy policies to accelerate change in its energy structure.

Also through scenario analysis, Guan and Barker [28] have used CO_2 emissions as an environmental indicator to investigate Guangyuan's future carbon performance and an alternative low-carbon development path. Technological improvements and production structure changes, which are interdependent, have been identified as the key determining factors likely to affect both carbon intensity and CO_2 emissions in the future. The paper concludes that governmental policies should include appropriate guidelines to address both factors, but with strong emphasis on the decarbonization of the production structure to avoid the mistake of 'pollute first and address the pollution later', which occurs during the emission-intensive industrialization processes that many Western countries and China's coastal regions have followed.

It is not easy to conceive of viable ways for China' cities to achieve cleaner and ecologically sustainable economic growth. Therefore, lowcarbon city development is often combined with the goal of developing knowledge cities. Because green/clean technologies are often believed to boost both the environment and economic development, importing

⁶ Ecological civilization is a concept proposed by Hu Jintao in his report to the concluded 17th National Congress of the Communist Party of China (in 2007). The concept recognizes the importance of harmonious development among humans, nature, and society, rather than solely focusing on economic growth as the core of development. Although it is unclear to what extent this concept guides policy processes, it signals a discursive shift in the official party line.

knowledge, expertise, science, and technology from foreign countries is considered to be a key component of developing low-carbon cities. Some researchers have suggested that it is useful, in terms of the lowcarbon development of China's cities, to learn from and collaborate with successful low-carbon cities in developed countries. Aiming to identify conditions for robust Sino-foreign partnerships in eco-cities, de Jong et al. [12] have examined how Sino-foreign initiatives are organized (including their strengths and weaknesses) and have constructed a typology of Sino-foreign initiatives that have focused on ecological knowledge related to cities. They have identified conditions for robust Sino-foreign partnerships in low-carbon cities. They found three types of collaboration between China and foreign partners. The first one is the least intense. It resembles a client-provider/designer relationship and offers very limited project success. The second type offers a more structured basis for interaction through an intergovernmental agreement-based collaboration. The third type goes much further and includes a large-scale joint venture in which both sides monitor the progress and quality of the project. Jakutyte-Walangitang and Page [36] have described a low-carbon city action plan for China's low-carbon pilot cities. It is a Sino-Austrian cooperation that has been initiated between the Development and Reform Commission of Nanchang, one of the eight selected Low-Carbon Pilot Cities, and the Austrian Institute of Technology to develop a comprehensive set of Low-Carbon City Measures and a Low-Carbon City Action Plan, proposing specific technological and non-technological measures and concrete actions. The plan is capable of introducing important changes targeting improved energy efficiency and the reduction of CO₂ emissions in Nanchang. A team of experts has developed an integrated Low-Carbon City Action Plan, including sectors such as buildings, energy supply and consumption, industries, transportation, agriculture, and urban planning. Dienst et al. [15] have presented some outcomes of the Sino-German Low-Carbon Future Cities (LCFC) project, including analysis of the status quo and an assessment of the likely future development of Wuxi City.

These studies have demonstrated that, similar to cities throughout the world, China's cities face a dilemma between economic growth and decarbonization. Researchers are therefore looking for new ways to coordinate industry and eco-efficiency so that the low-carbon city strategy can be realized. Currently, some researchers believe that industrial symbiosis (IS) provides a way to achieve system-level innovation useful in pursuing sustainable urban development. IS is a system-level innovation designed to share service, utility, and byproduct resources among diverse industrial processes or actors to add value, reduce costs and improve the environment [125,144]. IS can change the traditional linear structure of resource utilization by closing loops (in terms of material/waste recycling, heat exchange, etc.); thus, the environmental efficiency of the entire system or network is enhanced. According to the philosophy of IS, the linkage between industries and urban communities may enhance the eco-efficiency of both [16]. Dong et al. are the main advocates of this idea. They have used Liuzhou and Jinan as case studies to calculate the CO₂ emissions reduction potential under IS projects [16,17]. Moreover, by using scenario analysis, Wang et al. [98] have accounted for the industrial park of Suzhou City and analyzed future trends.

Broadly speaking, after the questions of the status of cities' emissions and what factors cause them, many researchers have begun to discuss which path can lead China's cities to achieve a low-carbon mode. Policies need to be implemented in practice, and much research can still be done on implementation processes. We clearly identify a lack of studies addressing feasible ways of switching to a new path, given the challenges of governance and the messy reality of most city development processes in the real world. Summarizing the literature, there is a visible need for all-encompassing systemic change, which cannot easily be generated through incremental policy measures, regulation, and technological improvements. The goal of a low-carbon city implies system-level innovation, as suggested by some studies on Sino-foreign partnership projects in this area. Among the required systemic changes, the energy policies and production technology improvements offer many ways for practical improvements. Finally, the city layout plays an important role in the entire establishment process of low-carbon cities. Although it is impossible to change the layout of an existing city in the short term, some changes of urban forms and functional areas can help to ease the urban pressure and lead to a lower carbon city. For example, Beijing has begun to explore the idea of polycentric city development by building Tongzhou District as a sub-civic center. This measure shifts some administrative and financial activities away from the central zones and reduces the stress of housing, traffic, and energy supply, making Beijing closer to a low-carbon city [69].

3. Current research on low-carbon technologies and industries in China

Technological innovations for industry processes and structural changes in energy- and carbon-intensive industries are additional research topics in the context of low-carbon development in China that have attracted much attention in recent years.

Low-carbon development of industries is an important concept within programs for developing low-carbon cities (see above); however, it deserves special treatment as an issue of importance for the national economy as a whole. One line of reasoning emphasizes that industrial production is the core of Chinese economic development and that it contributes the most to China's emissions. Many energy-saving and carbon reduction policies are targeted at industrial production and certain energy intensive industrial sectors. For example, in 2013, China's Ministry of Industry and Information Technology published a report that compared the CO₂ emissions of the iron and steel industrial sector, nonferrous metal industrial sector, petrochemical industrial sector, chemical industrial sector, machinery industrial sector, and textile industrial sector with those in 2010. The report indicated that CO2 emissions per unit of industrial added value should be reduced by approximately 17-22% by the end of 2015. Improving energy efficiency is critical to the low-carbon development of industrial production. China is also on its way to building carbon emissions trading systems (ETS). In seven ETS pilots, the targets have mainly focused on the important energy-intensive industrial sectors. As a developing country, China is facing the dual pressures of economic output and carbon reduction. Balancing this relationship is critical to China's industrial low-carbon transition. We will examine both issues, industry development and technological innovations, independently and then draw conclusions.

3.1. Low-carbon development of Chinese industries

China's industries are extremely energy intensive and are primarily responsible for the country's GHG emissions. The phrase "made in China" is familiar to the residents of the United States and Europe. China is the largest exporter in the world. Exported goods include new electronics, clothing, fashion accessories, and even medical equipment. One reason for the rapid growth of China's CO₂ emissions in recent decades is Western consumption patterns. Liu et al. [66] have analyzed changes in the GHG emissions of 36 industries in China during the 1998-2005 period. They have also explored the main factors of industrial GHG emissions. The changes in industrial CO2 emissions have been described by carbon emissions coefficients of heat and electricity, energy intensity, industrial structural shift, industrial activity, and final fuel shift. The authors have found that industrial activity and energy intensity contributed the most to the industrial sectors' GHG emissions during the study period. Raw chemical or chemical material products, nonmetal mineral products, and smelting or pressing of ferrous metals should be among the top priorities for enhancing energy efficiency because they are the most energy intensive among

China's 36 industries. The factors contributing to industrial CO_2 emissions have been studied by many researchers. In summarizing the relevant research, we find similar results: not only for the entire industrial sector but also for some separate sub-sectors, industrial output is the main factor causing CO_2 emissions to increase. In addition to the study by Liu et al. mentioned above, research by Li et al. [52], Ouyang and Lin [83], and Shao et al. [85] has also suggested that the level of industrial activity is the largest contributor to the CO_2 emissions of the industrial sector and that the energy intensity effect is the most important factor in emissions reduction. Studies by Fan et al. [19], Lin et al. [58], and [100,107] have explored the determinants of increased CO_2 emissions in the petrochemical industry, food industry, and iron and steel industries, respectively. For these industries, the level of industrial activity, that is, economic output, is the major determinant of CO_2 emissions.

According to these results, we can deduce that there has not been an obvious decoupling of industrial output and GHG emissions because industrial output is by far the most important factor. This is further supported by Li et al. [52], who have tested the decoupling relationship between CO_2 emissions and economic growth in China, suggesting that there is only a weak decoupling between the two. Zhang and Da's [126] research also proves the weak decoupling relationship from 1996 to 2010, indicating that the economy grew with increased carbon emissions. Similarly, the results of Zhao et al. [131] show that the industrial emissions from 1993 to 2013 experienced weak decoupling with economic growth.

Thus, reduction policies in China are all based on carbon intensity instead of total emission amounts. Liu B et al. (2015) have analyzed the factors influencing carbon intensity in industrial production and indicate that the energy intensity effect is most important compared with the emission coefficient effect and the structural effect. This result suggests that low-carbon industrial development will still depend on energy efficiency improvements; as studies have shown, energy intensity is the main factor contributing to emission reductions. Lin and Long [59] have made an interesting innovation, finding that the output per worker (OW) is an important factor that causes the increase of GHG emissions. The OW is formulated as industrial outputs for every worker. With the development of advanced technologies, labor has been gradually replaced by equipment and capital. The larger output per worker means more energy consumed by more equipment, leading to more GHG emissions [59].

So far, we have summarized findings about industry in general. What can we see if we differentiate between different industry sectors?

The iron and steel industry is one of the major industrial sectors contributing to energy consumption and CO₂ emissions in China. The GHG emissions of this industry mainly come from coal combustion, and thus, the increase in energy demand has exerted great pressure on the availability of energy resources and the goal of GHG emissions reductions. Sun et al. [92] have described the total CO₂ emissions change from 1980 to 2008 using four factors: emission coefficient, energy structure, energy consumption, and steel production output. The results show that production output is the major factor responsible for the increase in CO₂ emissions. In addition, shorter time series (1995-2007) have been used to analyze the influencing factors of energy-related CO₂ emissions in the iron and steel industries [93]. Lin and Wang [60] have conducted an intensive analysis of CO₂ emissions efficiency and carbon mitigation potential, suggesting that the iron and steel industry should put more effort into adopting energy-saving techniques and enhancing cooperation. However, from a long-term perspective, the fundamental solution to low-carbon development lies in energy substitution. Regarding the low-carbon development of the iron and steel industry, much work has been conducted from the macro perspective. Zhang et al., however, have conducted an empirical analysis of 85 questionnaires regarding various companies in the iron and steel industry to explore the determinants (regulatory pressures, pressures from the supply chain, environmental practices of competitors, financial cost, etc.) that drive or hinder the implementation of low-carbon practices [124]. The results suggest that regulatory policies promote the development of CO_2 reduction strategies at the management level but have no significant influence on energy-saving practices and CO_2 reduction in the actual production processes. Because most regulations imply only voluntary commitments to reducing carbon emissions, they fail to have significant effects on detailed practices. However, regulations do direct companies to start analyzing their CO_2 emission problems. Pressure from the supply chain may also produce positive effects on CO_2 reduction practices, whereas the lack of financing for implementing CO_2 reduction strategies through technological changes is a strong barrier.

Another important industry is the *building industry*. In China, the building industry has become an increasingly important energydemand sector, accounting for more than 20% of final energy consumption [49]. The design and construction of urban infrastructure with a long lifespan - of buildings in particular - has the potential to shape energy perspectives for the coming decades. Therefore, reducing the GHG emissions of the building industry is very important not only for the development of a low-carbon economy but also for the construction of low-carbon cities. Li has conducted a review of the literature regarding the forecasting of GHG emissions to investigate the energy-saving and GHG emission mitigation potential offered by the implementation of energy efficiency policies in the building sector [48]. Later, Li and Colombier assessed the role of energy efficiency in buildings for addressing climate change mitigation [49]. These two works provide a comprehensive overview of the characteristics, energy consumption status and development trends of this industry. They also outline economic and institutional barriers to large-scale deployment of sustainable, low-carbon, and carbon-free construction techniques, including high information costs, the persistence of carbon-intensive building techniques, the lack of appropriate incentives for landlords, and limited access to financing and energy subsidies. Furthermore, they cite as institutional barriers the fragmentation of the building industry and of the design process into many professions, trades, work stages, and industries. Although China issued a comprehensive multisectoral energy conservation plan over the 2006-2010 period, the authors note the resistance of actors in the building sector to this plan and its low enforcement and implementation rate in many medium and small cities because of inadequate technical and institutional capacities. In their opinion, cost effectiveness and policy relevance should be considered as the primary criteria in assessing mitigation programs in the building sector [49]. Zhang and Zhou [128] have analyzed the relationship among emission reduction regulations of governments, emission reduction awareness of companies, and their practical behaviors. The findings reveal that the related regulations, including direct regulations and economic carbon instruments, can increase companies' real improvement in emission reduction awareness and behaviors, indicating that for policy makers, adopting carbon reduction regulations is an effective way to encourage companies to consider more advanced carbon management [128]. Chen et al. [6] have presented a low-carbon building evaluation framework that can be used to monitor the effects of such regulations. They have accounted for the GHG emissions of buildings by dividing the life cycle of buildings into nine stages: building construction, fitment, outdoor facility construction, transportation, operation, waste treatment, property management, demolition, and disposal. It is argued that some economic and policy instruments can be used to foster the transition of this life cycle to low-carbon development. These instruments could include energy labeling and certification, establishing internal carbon markets, and encouraging energy pricing reforms.

Wang et al. [103] and Hu et al. [32] have analyzed low-carbon production in the Chinese *cement industry* at the macro and micro levels. Wang Y et al. have presented a GHG emission inventory of the cement industry in China and identified the main driving factors influencing GHG emission changes in the cement industry. Hu and colleagues selected two cement companies with different production processes to investigate material/energy use as well as emissions. The shaft kiln and the NSP (New Suspension Preheater) kiln are the two most common production processes in China. The two plants investigated in this paper have either a shaft kiln production line or an NSP kiln line. The study results show that the efficiency of energy consumption (including electric power consumption and overall energy consumption) is lower for the shaft kiln than for the NSP kiln. As for environmental emissions, the shaft kiln emits more SO₂, CO2₂, NO_x, and dust per unit of product than the NSP kiln and has higher ecoenvironmental impacts. Shen et al. [86] have also quantified the CO₂ emissions of China's cement industry based on the Life Cycle Assessment method.

Others have looked at energy efficiency in various Chinese industries comparatively. Xia et al. have conducted an integrative assessment of the energy efficiency of 38 industries in China [111] and found that the industries with the lowest energy efficiency are smelting and pressing of ferrous metals, manufacture of raw chemical materials and chemical products, and manufacture of non-metallic mineral products. In 2012, Wu et al. constructed both static and dynamic energy efficiency performance indexes for measuring industrial energy efficiency performance with CO_2 emissions [109]. Their study shows that the most important potential for energy efficiency improvements in China's industrial sector mainly comes from technological improvements.

In summary, industrial production is a major GHG emitter in China. Of all industries, the steel and iron industry is among the most important for its high energy intensity. The main drivers of industrial GHG emissions are industrial output and energy intensity. There are effective ways for industries to enhance their energy efficiency from a short-term perspective, including adopting energy-saving techniques and importing advanced technologies from developed countries. Consequently, their energy intensity can be reduced, but so far, no real decoupling between output growth and carbon emissions has been achieved. Therefore, carbon reduction policies based on total emission amounts will still be in conflict with the growth targets of the industries: as long as industrial output is the most important factor, output cannot be increased without also increasing carbon emissions. Therefore, the most important option lies in eliminating inefficient production technology and improving energy efficiency by widely applying energy-efficient technologies in the secondary industry, while from a long-term perspective, a shift in the energy mix is an even more important step toward low-carbon development. Achieving real lowcarbon development will be difficult unless China ends the highly coaldependent development mode of its industries.

3.2. Development of low-carbon technologies in China

It is generally acknowledged that the Chinese economy is coal dependent. Coal, the most abundant energy resource, is expected to dominate China's energy scheme for a very long time [8]. Thus, the development of clean coal technologies is crucial to China meeting its low-carbon targets. In 2005, Glomsrød and Wei [25] analyzed the question of whether using clean coal technologies is a viable strategy for reducing carbon emissions and improving China's environmental quality. Using a computable general equilibrium (CGE) of the Chinese economy, this work assesses the multiple benefits and costs of coal cleaning. Currently, the clean coal technologies being developed in China mainly include high efficiency combustion and advanced power generation technologies, carbon capture and storage (CCS) technologies, and integrated gasification combined cycle (IGCC) technologies. However, the development of low-carbon technologies in China is still at the initial stage. Chen and Xu have described the role of coal in the Chinese energy system and the recent achievements of Chinese clean coal technologies [8]. CCS may have been seen as a solution for resolving environmental dilemmas, addressing energy security issues,

and meeting low-carbon development targets. To date, CCS – including demonstration sites – has been a focus of research, but it has not been commercialized, and it is unlikely, given current trends, that this technology will find large-scale application either in China or elsewhere before 2030 [129]. Liu and Gallagher [64] have explored opportunities in Chinese CCS technology and have identified critical CCS-enabling technologies. Based on their results, they have proposed a roadmap for China that includes four steps [64], suggesting that in each step, major government support would be needed to implement a CCS strategy and achieve widespread commercialization of this technology.

Other studies have focused on low-carbon technologies in individual industries. For example, focusing on the coal chemical industry, Xie et al. have considered certain key technologies, including direct and indirect coal liquefaction, large-scale and high-efficiency coal gasification technology, the exploitation of coal poly-generation systems, and large-scale production of methanol and dimethyl ether (DME) technology using coal and COG [113]. Fischer [21] has analyzed the barriers to and challenges of low-carbon technology diffusion in China's photovoltaic industry. When developing low-carbon technologies, the role of companies and their R & D activities are also important. Liu and Gao [73] also discuss the issue of low-carbon technology diffusion but focus on the iron and steel industry. Using data from 60 sample companies in the iron and steel industry, the study argues that a moderate carbon price may generate a relatively more significant impact on the diffusion of low-carbon technologies compared to non-price related policies. A carbon price may drive companies to improve their carbon management, leading to a wider use of low-carbon technologies [73].

Zhou et al. [139] have discussed the role of joint R & D in enhancing companies' innovation capacity. The findings indicate that joint R & D has improved Chinese companies' technological capacities, human resources, and financial growth potential. Lin and Yang [61] have studied the role of investment in achieving efficiency gains in China's power industry. The results show that there exists a relatively vast space in which to promote the efficiency of the power industry. Furthermore, the investment structure reform of the power industry has the potential to provide significant impetus to improve efficiency. The authors conclude that foreign capital should be positively introduced while the proportion of state-owned assets is lowered to break the state's monopoly. To summarize, many technological options currently under discussion center around providing cleaner coal, as coal is expected to stay dominant in the Chinese energy structure for some time. While several technologies are available, the research reveals a low level investment in these technologies and a lack of incentives through carbon pricing as major challenges.

4. Current research on the transition of China's energy system

Cities and the industrial production sector are two major sites where low-carbon developments can be fostered. However, a look at China's total energy demand and at the structure of the entire energy system can provide more comprehensive insights. Which changes in the Chinese energy demand are expected in the coming decades, how much does the structure of the current energy system contribute to carbon emissions, and what are the prospects of switching to a lowcarbon energy system in China?

Many studies concerning the forecasting of energy consumption and GHG emissions in China from both short-term (until 2020) and mid- to long-term (2030, 2050) perspectives exist. Scenario analysis is a popular method for forecasting and typically includes low-carbon electricity scenarios, implementation of low-carbon policies, and technological development. At the mid- and long-term levels, Chai and Zhang [5] and Zhou et al. [135] have forecasted primary energy consumption trends for 2020, 2030, and 2050 and have made comparisons among them. Similar research included the predictions of energy consumption until 2030 [33] and of energy consumption, CO_2 emissions, and carbon intensity until 2050 [138,7]. The outcome of these scenarios is that in the long term, the economic structure will have been well adjusted, with much improved efficiency, leading to a decrease in the growth rate of carbon emissions. After 2030, along with a population stabilization scenario and an optimization of the industrial and energy structures, the amount of GHG emissions will start to decrease in absolute terms. Moreover, if more low-carbon technologies such as CCS can be applied, the peak time may occur earlier.

The scenarios differ in their assumptions about how much and how fast low-carbon changes can be achieved in the short term. On the one hand, some projections are based on scenario analysis in which both energy structure and economic structure will be readjusted toward a sharp decline in carbon intensity. However, the results show that even if the CO₂ intensity of GDP can be reduced by 50%, total CO₂ emissions in 2020 would still be the same as in 2005 [112,29]. The energy demand has also been forecasted by taking energy/economic structure improvement as variables [38], which shows that China's energy demand will maintain high growth until 2020, and the higher the economy grows, the larger the energy demand becomes. On the other hand, some researchers argue that in the short term, it is difficult to change China's energy-intensive economic structure. Moreover, the capacity for improving energy efficiency is limited [136,47]. Such studies suggest that encouraging the use of non-fossil energy or lowcarbon energy, especially the use of hydro-, wind, and nuclear power, is necessary for the Chinese government to achieve the emission reduction targets. Liu et al. [68] indicate that the mitigation potentials of renewable energy are much larger than those of energy efficiency and CCS. A strategy toward non-fossil-based energy sources could also have the much-wanted side effect of increased energy security, as it would ease the pressure to import oil to China [43,79]. We will now look in more depth at the two specific issues of changes in the energy demand (4.1) and the switch of the Chinese energy system toward non-fossil energy sources (4.2).

4.1. Changes in the energy demand

Some authors have analyzed the structure and development of Chinese energy demand at a systemic level and how these relate to carbon emissions. He et al. [29] have identified the main features of GHG emissions from fossil energy combustion in China as a whole, showing large annual CO2 emissions with rapid increases, low levels of per capita CO₂ emissions, high energy intensity, and a large CO₂ emissions factor due to China's coal-dominated energy structure. In a recent study, Lin and Ouyang [63] have not only analyzed the characteristics of energy demand in China but also compared these characteristics with the situation in the United States. They have shown that economic growth, urbanization, and industrialization are the leading forces contributing to the increase in energy demand, whereas improved energy intensity, advances in technology and increased energy prices contribute to a decline in energy consumption. The energy consumption of both the United States and China increased significantly during their urbanization stages. For energy consumption per capita, there are similarities and differences between the two countries. The common feature is that the energy consumption per capita of each country showed an upward trend in the rapid urbanization stage, with the difference that the energy demand per capita of China was far below that of the United States. Finally, for the energy consumption structure, coal was dominant in the energy structure during urbanization in both countries. However, coal has gradually been replaced by petroleum and natural gas in the United States, whereas China's coal-dominant energy structure has not yet shown any signs of shrinking. China's high energy consumption and GHG emissions are comprehensive results of its rapid urbanization and industrial output growth. For many years, the stable long-term correlations between population growth, economic growth, the urbanization process, and energy consumption remained unbroken [119,38,75], but the system is now under stress for various reasons, including fast-growing demand and structural change, growing environmental awareness and policies, and the rising costs of generation, transition, and distribution [40].

Following this observation, one important question would be at which level of decision-making, and how, the energy demand can be managed effectively: at the level of the central government and planning, at the provincial or lower levels, at the household levels, or within individual companies. Kahrl et al. [40] have argued that in the Chinese electricity system, the processes of planning, project approval and rate setting (i.e., determining the price of electricity) are still performed by the central government. The authors discuss two different ways to approach these problems: a top-down and a bottom-up approach. The top-down process would establish a legal framework for electricity sector jurisdiction and decision-making. In this case, electricity-generating companies would passively react to the legal framework. In contrast, the bottom-up process would rely on active management decisions at the company level. This would require the build-up of planning capacities and the development of analytical tools that could support a more cost-reflective electricity system and improve regional management. The authors emphasize the importance of management decisions at the company level and thus argue for strengthening the bottom-up approach, which would help to maximize benefits and minimize costs, both systematically and transparently. In their study on an optimal CO₂ mitigation strategy, Liu et al. [76] have also taken into consideration certain management issues in energy systems, e.g., how to cope with uncertainties and complexities related to carbon emissions. Under the assumption of stochastic power generation demand under specific GHG emission reduction targets, uncertainties will stem from the levels of power demand influenced by weather changes, population growth, and economic growth. According to the authors, current thinking about allocation strategies for CO₂ emission permits does not reflect these fluctuations and tendencies. Decision alternatives, including expanding generator capacity, switching to a more energy-efficient facility, coal selection, and coal blending, could be identified as desired strategies under environmental constraints. Li et al. [45] have studied energy system management to address interactions among energy supply, processing and demand activities, and associated GHG emissions, in particular the interactions with changes in the energy demand at the household level.

The household level has become a hot topic in recent years and includes both direct energy consumption and GHG emissions [26,27] and indirect types of material consumption and emissions [105,118,141,143]. In a more systematic reflection, the household sector can be divided into urban households [116,133,27] and rural households [118,57,70]. The indirect carbon emissions from "indirect energy requirements" are defined as the carbon emissions from the production of goods that are ultimately consumed by households. The indirect energy demand is usually not obvious, as it has a complicated relationship with the process of producing goods. The results from most studies have indicated that the material consumption level, emission intensity of consumed products, urbanization level, industrial structure (as discussed above), population, and consumption structure (the proportion of low-carbon products to the total consumed products) are the main factors influencing the energy consumption and GHG emissions of the household sector [104,105,10,67]. Among these factors, the material consumption level, which is usually measured by household consumption per capita, plays a dominant role in energy consumption and GHG emissions growth, whereas declining carbon intensity of the consumed products may mitigate such growth [105,118,67]. According to the forecast of Dai et al. [10], as incomes increase in the coming decades, energy consumption and GHG emissions in China's household sector will continue to rise dramatically. Therefore, changing patterns of material consumption to less carbonintensive products and to service-oriented goods may be a strategy for saving energy and reducing emissions at the household level [10,67].

To summarize, many studies note the complex causal linkages between different levels of the production and consumption of goods and services in the development of the total energy demand, and most studies do not expect the energy demand to decrease any time soon.

4.2. Switching to non-fossil energy sources

Switching from an economy based on fossil fuels to an economy (at least partially) based on renewable and low-carbon (cleaner) energy sources has been identified as the key to a low-carbon energy transition [79]. So far, China is heavily dependent on high-carbon fossil fuels. The long-standing dominance of coal in the energy mix has caused severe air pollution, water pollution, and loss of land resources and has thus been a major threat to China's environmental sustainability [5]. Researching issues of GHG emissions reduction and low-carbon transitions within energy systems, especially the Chinese electricity system, is critical. From the perspective of domestic low-carbon development, the Chinese electricity system has characteristics of high energy consumption, high carbon emissions, and high pollution. The low-carbon transition and clean production of power may play an important role in China's sustainable development. However, being predominantly coal based, the Chinese electricity system is also the world's single largest source of CO2 emissions [35]. Coal-fired power accounts for approximately 80% of total electricity generation, and emissions from coal-fired power plants account for approximately 40% of total emissions in China [114,134]. Therefore, from a global perspective, the transition to a low-carbon electricity system in China is critical to global efforts to combat climate change [40].

The Chinese government has designed a series of fundamental strategies for energy development that promote a partial replacement of fossil fuels but remain far from a substantial switch in the foreseeable future. These strategies include low-carbon or clean energy development, favoring energy conservation over further expansion, exploiting hydropower in accordance with local conditions, and developing nuclear and renewable energy [37]. For China, a country with abundant renewable energy resources, development of these resources may not only ensure energy security, improve the energy structure, and reduce negative effects on the environment but also alleviate rural energy poverty. Energy poverty in China, characterized by fuel poverty and power shortages, occurs at the household level and is worse in rural households. Over 75% of rural households use solid fuel such as wood or weed, whereas these figures are much lower in urban areas and townships, i.e., 8% and 36%, respectively [94]. The large amount of solid fuel usage in rural areas is closely tied to low household incomes [101], thus explaining why it is considered to be more important to develop non-fossil energy in China compared with some developed countries.

Some studies have summarized the current situation of renewable energy development in China. They have detailed the characteristics of China's energy structure and analyzed current issues in energy systems. For example, Shi [87] has provided an overview of the policy measures implemented over the past decade to support renewable energy development and has analyzed the current characteristics and conditions of renewable energy development. The implementation results of development goals and policy measures are also considered. Jiang et al. [37] have analyzed the pressure and challenge of energy development, focusing on the status of Chinese hydro-, wind, and nuclear power.

The potential contribution of renewable energies to achieving the goal of GHG emissions reduction has also been studied. Zhou et al. [136] have explored the development of non-fossil energy in China until 2020 and examined the role of renewable energy in achieving mitigation targets. Aiming to provide a scientific basis on which governments can set countermeasures against climate change and establish low-carbon energy development strategies, Hu et al. [33] have concentrated on the potential and contribution of low-carbon energy for achieving China's carbon intensity reduction goals. Ren and

Sovacool [84] have assessed the effect of renewable energies by determining the relative performance of hydropower, wind energy, solar energy, biomass energy, and nuclear power with respect to the energy security dimensions of availability, affordability, accessibility, and acceptability. They conclude that hydropower and wind power are the most promising low-carbon energy sources for enhancing China's energy security, whereas nuclear power and solar energy have the least potential. Liu et al. [68] have looked at the actual development status of renewable energy. Moreover, an analysis of CO₂ mitigation costs, mitigation potential, and the fossil energy conservation capacity of renewable energy has been conducted, resulting in a high mitigation potential of renewable energy. According to the authors, among the renewable energy resources, hydropower and wind should be prioritized because of their relatively more mature technologies and lower prices. Electricity from renewable energy will save a substantial amount of traditional fossil fuels, and its cost tends to decrease gradually. Lin et al. [62] have investigated the factors influencing the share of renewable electricity in the energy consumption structure. The results indicate that economic development and financial development have positive impacts, while foreign direct investment has negative impacts. This would suggest a more strategic approach to foreign direct towards the development of renewable electricity [62].

Some research has focused on individual renewable energy sources in China. For example, Huang and Yan [34] have examined the development status of *hydropower* in China, as well as factors restricting hydropower development. For *wind power*, Liu and Kokko [77] have discussed the recent development of the wind power industry, focusing on pricing policies, transmission capacity and the structure of the equipment manufacturing industry. They have also considered policies and challenges related to the wind power business environment.

The development of nuclear power may be more complicated. Because of rapid growth in electricity demand, and adjustment of its power structure, China has accelerated construction of nuclear power plants. Considering the increasing cost of oil and natural gas, and the enormous environmental pressure resulting from coal consumption, nuclear energy is considered to be an inevitable strategic option for China [137]. Nuclear energy is regarded as a sustainable option in the Chinese context because it can replace fossil energy on a large scale. Studies have suggested the possibility of technological improvements for nuclear safety and for substantial cost reductions. The unsolved problem of long-term nuclear waste treatment and storage is sometimes also discussed [137] but does not play a prominent role in current policy developments. China's government has announced its intention to speed up nuclear energy development. The Medium- and Long-term Nuclear Power Development Plan was first released in 2007. Thereafter, the successive plans associated with the nuclear power development were made. China's nuclear generation capacity is planned to increase to 40 GW by 2020, with a further 18 GW under construction, which means that nuclear power construction has entered a stage of accelerated development [80]. After the Fukushima accident in 2011, the Chinese government has increased its attention to nuclear safety. The related laws and policies are applied to each stage of nuclear power development, including site selection, construction, commissioning, operation and decommissioning [122,80]. The accident also had negative impacts on the public acceptance of nuclear energy. Zhu et al. [140] tested the public acceptance change by using the data of land price near nuclear power plants in China. The results show that the land prices have dropped within areas 40 km outside nuclear power plants, especially, the price dropped by an average of 18% in the first month after the Fukushima accident. This reflects the people's repulsion of living near nuclear power plants, indicating a decrease of public acceptance.

After decades of effort, China has managed to establish a comprehensive energy industry system. Moreover, efforts have been made to enhance energy efficiency, improve industry as well as energy structure, and promote a low-carbon energy transition. However, there are still several issues remaining to be solved. China's current total energy efficiency is still 10% lower than that of developed countries [37]. This is mainly because of China's lack of effective energy management. Consequently, the percentage of energy consumption in China from low-efficiency equipment is almost three times greater than the average level in developed countries. The development and application of renewable energy requires not only political and institutional support, but economic and market incentives. Early research pointed out that both private and state-owned renewable energy production companies are overly reliant on support from the government. Although many strategies have been designed to promote the development of renewable energy, what is lacking are policies that consider the market mechanism and provide financing [87]. Until now, these problems are still being lively discussed. Authors identify a lack of legal framework in support of renewable energy [123], defining the rights, responsibilities and obligations for each actor in the renewable energy industry, in order to avoid "passing the back" phenomenon caused by confused or overlapped responsibility. Technology standards are another element which needs to be developed to support renewable energy development [121]. Moreover, many authors raise the issue of commercialization mechanisms. At the current stage, the development of the renewable energy industry in China depends heavily on government support, subsidies, and funds from Clean Development Mechanism (CDM) projects. Once these financial supports are withdrawn, the renewable energy industry is expected to shrink again. Authors thus suggest the gradual replacement of government programs by the market mechanism [110,132,46].

5. Research regarding low-carbon development in Chinese journals

This paper is based mostly on research published in mainstream English-language journals. However, there is also a debate on lowcarbon strategies in Chinese journals, which is difficult for Western observers to follow. Before we come to the concluding chapter, we therefore offer a brief summary of our findings of the literature in Chinese journals. A review of this research has shown that very similar topics have been discussed and that there are no systematic differences between these bodies of literature. However, a few points are emphasized in Chinese journals that rarely appear in English journals. Taking low-carbon cities as an example, there is less dominance of mathematical modeling and analysis in Chinese journals compared to English journals. First, Chinese articles treat in more detail the concept of lowcarbon cities (definitions, main features, and significance) [115,11,23,51,55,56,89]. According to this literature, low-carbon cities can be defined by five features. The first is economic, meaning that minimum input should be used to produce maximum output. Second, low-carbon cities should be secure, including economic security, ecological security, and social security. Next is systematic construction, meaning that the establishment of low-carbon cities should be a systematic and comprehensive project. Moreover, building low-carbon cities should be a dynamic and continuous process. Finally, low-carbon cities should have regional characteristics, meaning that the construction plan should take local features into consideration.

Another specific feature is the more detailed assessment of the practice of developing low-carbon cities, including comparing China's low-carbon cities with foreign cities. In a study by Liu and Wang, China's development model is compared with those of London, Tokyo, Copenhagen, and others. China's cities have not established a clear method of low-carbon development. In contrast, mature low-carbon cities have sufficient experience building low-carbon industries, which in turn support the development of the entire city in a low-carbon way. On the basis of the practice of Masdar (United Arab Emirates) and Malmö (Sweden), some Chinese cities, such as Shanghai, Chongqing, and Beijing, are trying to build low-carbon pilot zones to explore effective ways of building low-carbon cities [72]. Some researchers have tried to establish an evaluation indicator system to assess low-carbon cities in China [141,143,30,31].

Many parties are involved in low-carbon city construction, including governments, enterprises, research centers or universities, and citizens, but this is seldom discussed in English journals. In contrast, some studies in Chinese journals provide a normative suggestion of the respective roles of these different parties [55,56]. The most important task of governments is thus to establish a scientific plan for the city's development. They are also responsible for the diffusion of the lowcarbon concept and the promotion of low-carbon awareness. At the same time, governments must build an assessment index system and incentive mechanisms. Enterprises should encourage low-carbon production to provide low-carbon products to the market. The role of research centers is to focus on low-carbon technologies so that they can provide technological support for the development of low-carbon cities. Citizens should develop awareness of low-carbon consumption and lead low-carbon lifestyles [11,120].

Specific issues that arise during the development of low-carbon cities have also been identified in Chinese publications [115,72,89]. The development of a low-carbon city should not occur at the expense of the city's other development goals. This means that the low-carbon target is not the only indicator that should be evaluated. The establishment of low-carbon cities cannot be treated as simply an investment in power generation or other infrastructure. It is always a long-term project. Low-carbon establishments should be designed that take local features into consideration. Problems with the current construction of low-carbon cities are also analyzed by studies in Chinese journals. For example, the goals of governments in building low-carbon cities are not sufficiently clear. Some projects are carried out only to demonstrate the achievements of local officials. Furthermore, practices and studies mainly focus on large cities, whereas medium-sized or small cities are seldom mentioned [53]. Finally, actual housing demand is often overestimated, resulting in an unreasonable investment scale.

Some analyses of Chinese publications also refer to low-carbon technologies and industries. The government's role in the development of low-carbon technologies and industries is discussed extensively. Li has suggested that governments have the responsibility to support lowcarbon developments. First, governments should design laws and regulations to provide legal insurance for the development of lowcarbon industries. Second, governments are thought to have the responsibility to raise the environmental awareness of the public to promote the successful implementation of low-carbon practices. Third, governments should set comprehensive and scientific environmental targets to measure and assess emission reduction achievements. Fourth, governments should encourage enterprises to focus on innovative low-carbon technologies. Fifth, governments should aim to improve industrial structures and energy structures. Sixth, governments should guide cooperation with developed countries [51].

To summarize, even though there are broad similarities between the publications in Chinese and in English, there seems to be more debate in Chinese journals about the role of different parties in the various implementation processes, in particular the role of the government.

6. Identifying fields of action and research gaps

China has taken important steps toward reducing future carbon emissions, and a lively academic debate regarding how the country can achieve low-carbon development exists. In this paper, we summarize the most important findings and identify research gaps in the current literature on low-carbon development in China. The final chapter will first present some overall comments and then summarize findings and research gaps for the three topics of low-carbon cities, low-carbon technologies and industries, and the transition of China's energy system individually. Overall, the most important finding repeated in all chapters with some variation is that, so far, no effective decoupling between industrial output and carbon emissions has been achieved. Despite all gains in energy and carbon efficiency and despite the massive expansion of renewable energies, economic growth so far still unavoidably means rising GHG emissions. Therefore, for China, carbon

reduction policies are more likely to be based on carbon intensity instead of total emission amounts. This is in line with the Chinese official climate policy goal to achieve a peak in its absolute carbon emissions only in 2030.

Some of the debates about the most promising ways to achieve lowcarbon development address the role of the state and the effects of pricing and market mechanisms. Several studies have emphasized that the diffusion of low-carbon technologies or a further increase of renewable energies would depend on cost-avoiding behavior on the side of emitters and on new investments independent of the state. Typically, authors suggest market mechanisms and a (moderate) prize for carbon emissions. This resonates with the current trends in economic thinking about mitigation policies. However, the question of how market mechanisms work in the context of the current Chinese economic system, still heavily influenced by state funding and central planning, deserves more attention.

Our overview reveals the need for deep structural change if lowcarbon development is the short- and long-term goal. We find evidence in the literature that a shift in the industry structure and a switch of the energy system away from fossil fuels would be the most important routes to low-carbon development. Against this backdrop, both mainstream Anglophone journals and Chinese journals pay most attention to cities and their role in low-carbon development. Cities bring together many structural aspects for research on low-carbon development, as their industry and energy structures and, in general, the city layout and its major infrastructures greatly influence their carbon emissions and the ways cities can decarbonize. Judging from our literature overview, we conclude that cities have great potential for achieving the switch to a low-carbon development mode. However, low-carbon cities are still in the early phase, and the literature discusses many unresolved problems – e.g., of governance, of cooperation, of planning, and of funding.

A crosscutting issue in all three topics was China in the international context: how do China's strategies toward low-carbon development compare to strategies in different countries, and how can collaboration with foreign partners contribute to implementing the strategies? Comparison with strategies in other countries, e.g., at the city level, was shown to be difficult because of data problems, incoherent methodologies, and a lack of continuous monitoring. The effects of international collaboration always depend on the circumstances in which collaboration occurs: partner projects between cities can be anything from loose client-provider relationships to large-scale joint venture projects with effective long-term engagement on both sides. The effects of FDI were also tested in several studies, with inconclusive results - the potential carbon effects of FDI were shown to be both positive and negative. Studies would need to differentiate better between different types and sectors of FDI to improve our understanding of the potential contributions of international collaboration to low-carbon development.

One other issue became obvious in all preceding chapters: research on low-carbon development is characterized by a large diversity of methodologies, often with incompatible epistemological starting points and underpinnings. Explaining in detail how the methodologies we found would relate to each other, what types of knowledge claims they generate, and what their respective validity and explanatory model is was beyond the scope of our paper. Our overview led us to believe that much could be gained from mapping these methodological approaches systematically to better qualify the implications of each methodology's findings. However, we found very few papers that discuss their findings in light of findings generated by competing or even incompatible approaches. We suggest a more systematic comparison and evaluation in this regard and more comprehensive meta-studies.

After these general and cross-cutting comments, we will as our last point summarize our main suggestions for the three topics on which this paper focused:

Low-carbon cities

Few articles have presented a clear strategy or roadmap for establishing a low-carbon city. From a summary of the literature, however, we deduce some useful steps. First, the features of energy consumption and GHG emissions should be analyzed with tools that reflect local characteristics. Related environmental policies or goals set by local governments should also be considered. Moreover, a set of viable or proposed measures should be designed. During this phase, the key sectors should also be identified. Measures can be clustered by sectors or groups in both technological and nontechnological categories. However, each of them should be specific and have one clear target focusing on one industry. Next, a clear timeline should be considered. A careful weighing and prioritization of specific measures in each sector is required as part of this timeline. Time points for checking and monitoring should also be determined. Low-carbon cities can be better established through a general shift from multiple isolated measures (addressing individual technologies or sectors) to a more integrated approach (combining key carbon activities together). During the establishment process, the relationship between the local government and foreign partners must be carefully considered. A solid framework of bi-national political support, active public and private investment, and effective advice and communication between the two sides can contribute to successful collaboration initiatives [12]. Further research could address the following topics:

- Continuous inventory research for a city: not just for one year, but for the long term (continuous observation of emission changes in the city);
- (2) Assessment of low-carbon practices in a city, especially in China's low-carbon city pilot projects, focusing on how lowcarbon development goals are actually implemented;
- The role of governments and issues of governance during the construction of low-carbon cities;
- (4) A more elaborate social science analysis of local governance and urban planning. Most papers, especially papers in English journals, are mainly based on formal mathematical modeling. The societal and institutional preconditions of low-carbon city development in China are substantially under-researched.
- <u>Industry</u>

There is very little current literature on how management strategies can improve energy efficiency, promote the shift of the energy mix and generate more green investments. More in-depth studies of individual industries are required to complement the broad overviews and to identify at a more concrete level the technologies that can promote low-carbon production pathways under the conditions of specific industries. Identifying how management strategies can support the transition over the short and long terms is also critical. In addition, studies have shown that almost no decoupling of industrial activity and CO_2 emissions has been achieved so far and that enormous potential from technological improvements remains. Further research may address the following topics:

- The role of governments/enterprises/citizens in the development of low-carbon industries;
- (2) The relevance of management methods or public environmental awareness as drivers of the development of low-carbon industries;
- (3) Quantitative sector and multi-sector analyses that could be complemented by case studies adopting a micro perspective:

which low-carbon technologies have been used, what are the effects, and which organizational features have fostered or hindered their implementation?

- (4) Cost analyses of low-carbon production processes for companies in different sectors;
- (5) Development of a roadmap or specific steps for improving the industrial structure, and an analysis of benefits and losses produced by such adjustment.
- Transition of the energy system

In summary, clean coal technologies are considered to be an appropriate way to balance CO_2 reduction and economic development in China. Of these technologies, CCS has attracted much attention from both researchers and enterprises, although it is still at the demonstration stage and has not been applied at a large scale. Studies suggest that an improved investment structure must also be enhanced by increasing the proportion of foreign investment instead of state-owned investment, but other studies show ambivalent results. Further research may address the following topics:

- The drivers and hindering factors in the diffusion of renewable energy application in China;
- (2) Deeper discussions of management strategies for technologies, including issues of selection, ways of updating current systems, and cost estimations, to better promote low-carbon production;
- (3) Study of the collaboration mechanisms among governments, enterprises and research centers;
- (4) A broader discussion about the long-term costs and the risks of nuclear energy, including the problem of nuclear waste management, which might support a balanced assessment of all longterm non-fossil options for the Chinese energy mix.

Acknowledgement

This study is supported by the Program National Natural Science Foundation of China (Reference No. 71173017, 71573016, 71521002); the Cluster of Excellence Integrated Climate System Analysis and Prediction (CliSAP; Reference No. DFG EXC 177) provided institutional support in Hamburg.

References

- Alam S, Fatima A, Butt MS. Sustainable development in Pakistan in the context of energy consumption demand and environmental degradation. J Asian Econ 2007;18:825–37.
- [2] Bi J, Zhang R, Wang H, Liu M, Wu Y. The benchmarks of carbon emissions and policy implications for China's cities: case of Nanjing. Energy Policy 2011;39(9):4785–94.
- [3] Bulkeley H, Betsill M. Rethinking sustainable cities: multilevel governance and the 'urban' politics of climate change. Environ Polit 2005;14(1):42–63.
- [4] Bulkeley H, Betsill M. Revisiting the urban politics of climate change. Environ Polit 2013;22(1):136–54.
- [5] Chai Q, Zhang X. Technologies and policies for the transition to a sustainable energy system in China. Energy 2010;35(10):3995–4002.
- [6] Chen GQ, Chen H, Chen ZM, Zhang B, Shao L, Guo S, Zhou SY, Jiang MM. Lowcarbon building assessment and multi-scale input–output analysis. Commun Nonlinear Sci Numer Simul 2011;16(1):583–95.
- [7] Chen W, Wu Z, He J, Gao P, Xu S. Carbon emission control strategies for China: a comparative study with partial and general equilibrium versions of the China MARKAL model. Energy 2007;32(1):59–72.
- [8] Chen W, Xu R. Clean coal technology development in China. Energy Policy 2010;38(5):2123–30.
- [9] Chong WHB, Guan D, Guthrie P. Comparative analysis of carbonization drivers in China's megacities. J Ind Ecol 2012;16(4):564–75.
- [10] Dai H, Masui T, Matsuoka Y, Fujimori S. The impacts of china's household consumption expenditure patterns on energy demand and carbon emissions towards 2050. Energy Policy 2012(50 special issue):736–50.
- [11] Dai YX. The necessity and governance model of developing low carbon city in China. China Popul, Resour Environ 2009;19(3):12–7, [in Chinese].
- [12] de Jong M, Yu C, Chen X, Wang D, Weijnen M. Developing robust organizational frameworks for Sino-foreign eco-cities: comparing Sino-Dutch Shenzhen Low Carbon City with other initiatives. J Clean Prod 2013(57):209–20.
- [13] de Jong M, et al. Ecocity development in China: addressing the policy implementation challenge. J Clean Prod 2016;134:31–41.

- [14] Dhakal S. Urban energy use and carbon emissions from cities in China and policy implications. Energy Policy 2009;37:4208–19.
- [15] Dienst C, Schneider C, Xia C, Saurat M, Fischer T, Vallentin D. On track to become a low carbon future city? First findings of the integrated status quo and trends assessment of the pilot city of Wuxi in China. Sustainability 2013;5(8):3224–43.
- [16] Dong L, Fujita T, Zhang H, Dai M, Fujii M, Ohnishi S, Liu Z. Promoting lowcarbon city through industrial symbiosis: a case in China by applying HPIMO model. Energy Policy 2013;61:864–73.
- [17] Dong L, Gu F, Fujita T, Hayashi Y, Gao J. Uncovering opportunity of low-carbon city promotion with industrial system innovation: case study on industrial symbiosis projects in China. Energy Policy 2014;65:388–97.
- [18] Engels A, Qin T, Sternfeld E. Carbon governance in China by the creation of a carbon market. Polit Carbon Mark 2015.
- [19] Fan T, Luo R, Xia H, Li X. Using Imdi method to analyze the influencing factors of carbon emissions in china's petrochemical industries. Nat Hazards 2015;75(2):319–32.
- [20] Feng YY, Chen SQ, Zhang LX. System dynamics modeling for urban energy consumption and CO₂ emissions: a case study of Beijing, China. Ecol Model 2013;252:44–52.
- [21] Fischer D. Challenges of low carbon technology diffusion: insights from shifts in China's photovoltaic industry development. Innov Dev 2012;2(1):131–46.
- [22] Fong WK, Matsumoto H, Lun YF. Application of system dynamics model as decision making tool in urban planning process toward stabilizing carbon dioxide emissions from cities. Build Environ 2009;44(7):1528–37.
- [23] Fu Y, Liu YJ, Wang YL. Evaluation method and supporting system of low carbon cities. China Popul, Resour Environ 2010;20(8):44-7, [in Chinese].
- [24] Gertz M. New research aims to create accurate account of a city's carbon burden. Environ Sci Technol 2009;43:8006–7.
- [25] Glomsrød S, Wei T. Coal cleaning: a viable strategy for reduced carbon emissions and improved environment in China?. Energy Policy 2005;33(4):525–42.
- [26] Golley J, Meng X. Income inequality and carbon dioxide emissions: the case of Chinese urban households. Energy Econ 2012;34(6):1864–72.
- [27] Gu ZH, Sun Q, Wennersten R. Impact of urban residences on energy consumption and carbon emissions: an investigation in Nanjing, China. Sustain Cities Soc 2013;7:52-61.
- [28] Guan D, Barker T. Low-carbon development in the least developed region: a case study of Guangyuan, Sichuan province, southwest China. Nat Hazards 2012:62(2):243-54.
- [29] He J, Deng J, Su M. CO₂ emissions from China's energy sector and strategy for its control. Energy 2010;35(11):4494–8.
- [30] He YQ, Weng J, Yuan YY. Evaluation for low carbon development in jiangxi province-based on factor analysis. Enterp Econ 2011;12:65-7, [in Chinese].
 [31] Hou WX, Gao JZ. Empirical evaluation of low carbon economy for taiyuan city-
- [31] Hou WX, Gao JZ. Empirical evaluation of low carbon economy for taiyuan city-based on factor analysis. Enterp Econ 2012;6:15–9, [in Chinese].
 [32] Hu D, Guo Z, Wang Z, Xiao Q. Metabolism analysis and eco-environmental impact
- [32] Hu D, Guo Z, Wang Z, Xiao Q. Metabolism analysis and eco-environmental impact assessment of two typical cement production systems in Chinese enterprises. Ecol Inform 2015;26:70–7.
- [33] Hu Z, Yuan J, Hu Z. Study on China's low carbon development in an Economy– Energy–Electricity–Environment framework. Energy Policy 2011(39.5):2596–605.
- [34] Huang H, Yan Z. Present situation and future prospect of hydropower in China. Renew Sustain Energy Rev 2009;13:1652-6.
- [35] International Energy Agency (IEA). World Energy Outlook 2009. Paris: OECD/ IEA; 2009.
- [36] Jakutyte-Walangitang D, Page J. A Low Carbon City Action Plan for one of China's Low Carbon Pilot Cities. EPJ Web Conf 2012;33:05002.
- [37] Jiang B, Sun Z, Liu M. China's energy development strategy under the low-carbon economy. Energy 2010(35.11):4257–64.
- [38] Jiang Z, Lin B. China's energy demand and its characteristics in the industrialization and urbanization process. Energy Policy 2012(49):608–15.
- [39] Jotzo F, Löschel A. Emissions trading in China: emerging experiences and international lessons. Energy Policy 2014(75):3–8.
- [40] Kahrl F, Williams J, Jianhua D, Junfeng H. Challenges to China's transition to a low carbon electricity system. Energy Policy 2011;39(7):4032–41.
- [41] Kennedy CA, Ramaswami A, Carney S, Dhakal S. Greenhouse gas emission baselines for global cities and metropolitan regions. Cities Clim Chang: Responding An Urgent Agenda 2011:15–54.
- [42] Kong B. Governing China's energy in the context of global governance. Glob Policy 2011;2:51–65.
- [43] Leung Guy CK, Cherp Aleh, Jewell Jessica, Wei Ji-Ming. Securitization of energy supply chains in China. Appl Energy 2014;123:316–26.
- [44] Li B, Liu X, Li Z. Using the STIRPAT model to explore the factors driving regional CO₂ emissions: a case of Tianjin, China. Nat Hazards 2015;76(3):1667–85.
- [45] Li GC, Huang GH, Lin QG, Zhang XD, Tan Q, Chen YM. Development of a GHGmitigation oriented inexact dynamic model for regional energy system management. Energy 2011;36(5):3388–98.
- [46] Li H, Guo S, Cui L, Yan J, Liu J, Wang B. Review of renewable energy industry in Beijing: development status, obstacles and proposals. Renew Sustain Energy Rev 2015;43:711–25.
- [47] Li HQ, Wang LM, Shen L, Chen FN. Study of the potential of low carbon energy development and its contribution to realize the reduction target of carbon intensity in China. Energy Policy 2012;41:393–401.
- [48] Li J. Towards a low-carbon future in China's building sector a review of energy and climate models forecast. Energy Policy 2008;36(5):1736–47.
- [49] Li J, Colombier M. Managing carbon emissions in China through building energy efficiency. J Environ Manag 2009;90(8):2436–47.

C. Wang et al.

Renewable and Sustainable Energy Reviews 81 (2018) 1350-1364

- [50] Li L, Chen C, Xie S, Huang C, Cheng Z, Wang H, et al. Energy demand and carbon emissions under different development scenarios for Shanghai, China. Energy Policy 2010;38(9):4797–807.
- [51] Li S. Analysis of principles for government developing low-carbon industry. China For Econ 2011;2(13):27–9, [in Chinese].
- [52] Li W, Sun S, Li H. Decomposing the decoupling relationship between energyrelated CO₂ emissions and economic growth in china. Nat Hazards 2015:1–21.
- [53] Li X, Liu Y. The current situations, problems and solutions of Chinese eco-cities development. China Acad J Electron Public House 2010(4):23–9, [in Chinese].
- [54] Li X, Xiao R, You Y. Dynamic change and influential factors of carbon footprint for energy consumption: a case study of Wuhan City, Hubei Province, China. Environ Eng Manag J 2014;13(6):1497–508.
- [55] Li YY, Zhao GL. A review on researches of low-carbon city construction in China. Ecol Econ 2015;31(2), [in Chinese].
- [56] Li ZF, Zheng YH. Research on the implementation mechanism of the low carbon city. Econ Geogr 2010;30(6):28–33, [in Chinese].
- [57] Liang L, Wu W, Lal R, Guo Y. Structural Change and Carbon Emission of Rural Household Energy Consumption in Huantai, Northern China. Renew Sustain Energy Rev 2010;28(8):767-76.
- [58] Lin B, Lei X, France N. Carbon emissions reduction in China's food industry. Energy Policy 2015;86:483-92.
- [59] Lin B, Long H. Emissions reduction in China's chemical industry- based on LMDI. Renew Sustain Energy Rev 2016(53):1348-55.
- [60] Lin B, Wang X. Carbon emissions from energy intensive industry in China: evidence from the iron & steel industry. Renew Sustain Energy Rev 2015;47:746-54.
- [61] Lin B, Yang L. Efficiency effect of changing investment structure on China's power industry. Renew Sustain Energy Rev 2014;39:403–11.
- [62] Lin B, Omoju OE, Okonkwo JU. Factors influencing renewable electricity consumption in China. Renew Sustain Energy Rev 2016(55):687–96.
- [63] Lin B, Ouyang X. Energy demand in China: comparison of characteristics between the US and China in rapid urbanization stage. Energy Convers Manag 2014;79:128–39.
- [64] Liu H, Gallagher KS. Catalyzing Strateg Transform A Low-Carbon Econ: A CCS Roadmap China 2010(38):59–74.
- [65] Liu J, Deng X. Impacts and mitigation of climate change on Chinese cities. Curr Opin Environ Sustain 2011;3(3):188–92.
- [66] Liu LC, Fan Y, Wu G, Wei YM. Using LMDI method to analyze the change of China's industrial CO₂ emissions from final fuel use: an empirical analysis. Energy Policy 2007;35(11):5892–900.
- [67] Liu LC, Wu G, Wang JN, Wei YM. China's carbon emissions from urban and rural households during 1992–2007. J Clean Prod 2011;19(15):1754–62.
- [68] Liu T, Xu G, Cai P, Tian LH, Huang QL. Development forecast of renewable energy power generation in China and its influence on GHG control strategy of the country. Renew Energy 2011;36:1284–92.
- [69] Liu W, Qin B. Lo-carbon city initiatives in China: a review from the policy paradigm perspective. Cities 2016;51:131–8.
- [70] Liu W, Spaargaren G, Heerink N, Mol APJ, Wang C. Energy consumption practices of rural households in north china: basic characteristics and potential for low carbon development. Energy Policy 2013;55(4):128–38.
- [71] Liu W, Wang C, Xie X, Mol AP, Chen J. Transition to a low-carbon city: lessons learned from Suzhou in China. Front Environ Sci Eng 2012;6(3):373–86.
- [72] Liu WL, Wang C. Practice and patterns of low carbon city development. China Popul, Resour Environ 2010;20(4):17–22, [in Chinese].
- [73] Liu X, Gao X. A survey analysis of low carbon technology diffusion in China's iron & steel industry. J Clean Prod 2016(129):99–101.
- [74] Liu X, Liu J. Measurement of low carbon economy efficiency with a three-stage data envelopment analysis: a comparison of the largest twenty CO₂ emitting countries. Int J Environ Res Public Health 2016;13:1116.
- [75] Liu Y. Exploring the relationship between urbanization and energy consumption in China using ARDL (autoregressive distributed lag) and FDM (factor decomposition model). Energy 2009;34.11:1846–54.
- [76] Liu Y, Huang G, Cai Y, Dong C. An inexact mix-integer two-stage linear programming model for supporting the management of a low-carbon energy system in China. Energies 2011;4(10):1657–86.
- [77] Liu Y, Kokko A. Wind power in China: policy and challenge. Energy Policy 2010(38):5520-9.
- [78] Liu Z, Liang S, Geng Y, Xue B, Xi F, Pan Y, Zhang T, Fujita T. Features, trajectories and driving forces for energy-related GHG emissions from Chinese mega cites: the case of Beijing, Tianjin, Shanghai and Chongqing. Energy 2012;37(1):245–54.
- [79] Mathews John A, Tan Hao. China's energy industrial revolution. Carbon Manag 2014;5(1):1–3.
- [80] Mu R, Zuo J, Yuan X. China's approach to nuclear safety from the perspective of policy and institutional system. Energy Policy 2015;76:161–72.
- [81] NDRC. National programs on coping with climate change. (Available at: http://www.ndrc.gov.cn/xwfb/t20070604_139486.htm); 2006.
- [82] NDRC. (Available at: (http://unfccc.int/files/meetings/cop_15/copenhagen_ accord/application/pdf/chinacphaccord_app2.pdf)); 2010.
- [83] Ouyang X, Lin B. An analysis of the driving forces of energy-related carbon dioxide emissions in china's industrial sector. Renew Sustain Energy Rev 2015;45:838–49.
- [84] Ren J, Sovacool BK. Prioritizing low-carbon energy sources to enhance China's energy security. Energy Convers Manag 2015;92:129–36.
- [85] Shao C, Guan Y, Wan Z, Guo C, Chu C, Ju M. Performance and decomposition analyses of carbon emissions from industrial energy consumption in Tianjin, China. J Clean Prod 2014;64(2):590–601.

- [86] Shen W, Cao L, Li Q, Zhang W, Wang G, Li C. Quantifying CO₂ emissions from China's cement industry. Renew Sustain Energy Rev 2015;50:1004–12.
- [87] Shi D. Analysis of China's renewable energy development under the current economic and technical circumstances. China World Econ 2009;17.2:94–109.
- [88] Song D, Su M, Jin Y, Chen B. Greenhouse gas emission accounting and management of low-carbon community. Sci World J 2012(2):613721.
- [89] Su MR, Chen B, Chen C, Yang ZF, Liang C, Wang J. Reflection on upsurge of lowcarbon cities in China: status quo, problems and trends. China Popul, Resour Environ 2012;3:48–55, [in Chinese].
- [90] Sugar L, Kennedy C, Leman E. Greenhouse gas emissions from Chinese cities. J Ind Ecol 2012;16(4):552–63.
- [91] Sternfeld E. Mod Fueled Coal: Chall Facing China Energy Sect Ind Ecol 2013;16(4):552-63.
- [92] Sun WQ, Cai JJ, Mao HJ, Guan DJ. Change in Carbon Dioxide (CO₂) emissions from energy use in China's Iron and steel industry. J Iron Steel Res, Int 2011;18(6):31-6.
- [93] Sun W, Cai J, Yu H, Dai L. Decomposition analysis of energy-related carbon dioxide emissions in the iron and steel industry in China. Front Environ Sci Eng 2012;6(2):265–70.
- [94] Tang X, Liao H. Energy poverty and solid fuels use in rural China: analysis based on national population census. Energy Sustain Dev 2014(23):122–9.
- [95] Tong K, Fang A, Boyer D, Hu Y, Cui S, Shi L, Kalmykova Y, Ramaswami A. Greenhouse gas emissions from key infrastructure sectors in larger and smaller Chinese cities: method development and benchmarking. Carbon Manag 2016;7(1-2):27-39.
- [96] UN Habitat . Cities and climate change: global report on human settlements. UN Habitat; 2011.
- [97] Urban F. Climate-Change Mitigation Revisited: low-carbon energy transitions for China and India. Dev Policy Rev 2009;27.6:693–715.
- [98] Wang H, Lei Y, Wang H, Liu M, Yang J, Bi J. Carbon reduction potentials of China's industrial parks: a case study of Suzhou Industry Park. Energy 2013;55:668-75.
- [99] Wang H, Wang Y, Wang H, Liu M, Zhang Y, Zhang R, et al. Mitigating greenhouse gas emissions from China's cities: case study of Suzhou. Energy Policy 2014;68:482–9.
- [100] Wang H, Zhang R, Liu M, Bi J. The carbon emissions of Chinese cities. Atmos Chem Phys 2012;12(14):6197–206.
- [101] Wang K, Wang YX, Li K, Wei YM. Energy poverty in China: an index based comprehensive evaluation. Renew Sustain Energy Rev 2015(47):308-23.
- [102] Wang S, Fang C, Wang Y, Huang Y, Ma H. Quantifying the relationship between urban development intensity and carbon dioxide emissions using a panel data analysis. Ecol Indic 2015;49:121–31.
- [103] Wang Y, Zhu Q, Geng Y. Trajectory and driving factors for GHG emissions in the Chinese cement industry. J Clean Prod 2013;53:252–60.
- [104] Wang Z, Liu W, Yin J. Driving forces of indirect carbon emissions from household consumption in china: an input–output decomposition analysis. Nat Hazards 2015;75(2):257–72.
- [105] Wang Z, Yang L. Indirect carbon emissions in household consumption: evidence from the urban and rural area in china. J Clean Prod 2014;78:94–103.
- [106] Wang Z, Yin F, Zhang Y, Zhang X. An empirical research on the influencing factors of regional CO₂ emissions: evidence from Beijing city, China. Appl Energy 2012;100:277–84.
- [107] Wang Z, Zhang B, Yin J. Determinants of the increased CO₂ emission and adaption strategy in chinese energy-intensive industry. Nat Hazards 2012;62(1):17–30.
- [108] World Bank. CO₂ emissions (metric tons per capita). (http://data.worldbank.org/ indicator/EN.ATM.CO2E.PC> [accessed on 5 August 2015]; 2015.
- [109] Wu F, Fan LW, Zhou P, Zhou DQ. Industrial energy efficiency with CO₂ emissions in China: a nonparametric analysis. Energy Policy 2012;49:164–72.
- [110] Wu G, Zeng M, Peng L, Liu X, Li B, Duan J. China's new energy development: status, constraints and reforms. Renew Sustain Energy Rev 2016;53:885–96.
- [111] Xia XH, Huang GT, Chen GQ, Zhang B, Chen ZM, Yang Q. Energy security, efficiency and carbon emission of Chinese industry. Energy Policy 2011;39(6):3520–8.
- [112] Xiang N, Xu F, Sha J. Simulation analysis of China's energy and industrial structure adjustment potential to achieve a low-carbon economy by 2020. Sustainability 2013(5.12):5081–99.
- [113] Xie K, Li W, Zhao W. Coal chemical industry and its sustainable development in China. Energy 2010;35(11):4349–55.
- [114] Xu G, Liang FF, Wu Y, Yang YP, Zhang K, Liu WY. A new proposed approach for future large-scale de-carbonization coal-fired power plants. Appl Therm Eng 2015;87:316–27.
- [115] Xu LH. Path Choices of Low-carbon City Construction in China. Fudan J (Social Sci) 2015(2):135–41, [in Chinese].
- [116] Xu X, Tan Y, Chen S, Yang G, Su W. Urban household carbon emission and contributing factors in the Yangtze river delta, China. PLoS One 2015;10(4).
- [117] Yin Y, Olsson AR, Håkansson M. The role of local governance and environmental policy integration in Swedish and Chinese eco-city development. J Clean Prod 2016;134:78–86.
- [118] Yuan B, Ren S, Chen X. The effects of urbanization, consumption ratio and consumption structure on residential indirect CO₂ emissions in China: a regional comparative analysis. Appl Energy 2015;140:94–106.
- [119] Yuan JH, Kang JG, Zhao CH, Hu ZG. Energy consumption and economic growth: evidence from China at both aggregated and disaggregated levels. Energy Econ 2008;30(6):3077–94.
- [120] Yuan XL, Zhong YY. The Practice and System Construction of China's low carbon

city. Urban Stud 2010;17(5):42-7, [in Chinese].

- [121] Zeng M, Li C, Zhou L. Progress and prospective on the police system of renewable energy in China. Renew Sustain Energy Rev 2013;20(4):36–44.
- [122] Zeng M, Liu Y, Ouyang S, Shi H, Li C. Nuclear energy in the Post-Fukushima Era: research on the developments of the Chinese and worldwide nuclear power industries. Renew Sustain Energy Rev 2016();58:147-56.
- [123] Zeng M, Yang Y, Wang L, Sun J. The power industry reform in China 2015: policies, evaluations and solutions. Renew Sustain Energy Rev 2016();57:94–110.
- [124] Zhang B, Wang Z, Yin J, Su L. CO₂ emission reduction within Chinese iron & steel industry: practices, determinants and performance. J Clean Prod 2012;33:167–78.
- [125] Zhang H, Liang D, Li H, Fujita T, Ohnishi S, Tang Q. Analysis of low-carbon industrial symbiosis technology for carbon mitigation in a Chinese iron/steel industrial park: a case study with carbon flow analysis. Energy Policy 2013;61(10):1400–11.
- [126] Zhang YJ, Da YB. The decomposition of energy-related carbon emission and its decoupling with economic growth in China. Renew Sustain Energy Rev 2015;41:1255–66.
- [127] Zhang L, Feng Y, Chen B. Alternative scenarios for the development of a lowcarbon city: a case study of Beijing, China. Energies 2011(12):2295–310.
- [128] Zhang L, Zhou J. The effect of carbon reduction regulations on contractors' awareness and behaviors in China's building sector. J Clean Prod 2016(113):93-101.
- [129] Zhang Z. China in the transition to a low-carbon economy. Energy Policy 2010;38(11):6638-53.
- [130] Zhao R, Huang X, Liu Y, Zhong T, Ding M, Chuai X. Carbon emission of regional land use and its decomposition analysis: case study of Nanjing City, China. Chin Geogr Sci 2015:1–15.
- [131] Zhao X, Zhang X, Shao S. Decoupling CO₂ emissions and industrial growth in China over 1993–2013: the role of investment. Energy Econ 2016;60:275–92.
- [132] Zhao ZY, Yan H, Zuo J, Tian YX, Zillante G. A critical review of factors affecting the wind power generation industry in China. Renew Sustain Energy Rev 2013;19(02):499–508.
- [133] Zheng S, Wang R, Glaeser EL, Kahn ME. The greenness of china: household carbon dioxide emissions and urban development. J Econ Geogr

2011;11(5):761-92, [32].

- [134] Zhou KL, Yang SL, Shen C, Ding S, Sun SP. Energy conservation and emission reduction of China's electric power industry. Renew Sustain Energy 2015;45:10-9.
- [135] Zhou N, Fridley D, Khanna NZ, Ke J, McNeil M, Levine M. China's energy and emissions outlook to 2050: perspectives from bottom-up energy end-use model. Energy Policy 2013;53:51–62.
- [136] Zhou S, Tong Q, Yu S, Wang Y, Chai Q, Zhang X. Role of non-fossil energy in meeting China's energy and climate target for 2020. Energy Policy 2012;51:14–9.
- [137] Zhou S, Zhang X. Nuclear energy development in China: a study of opportunities and challenges. Energy 2010;35.11:4282–8.
- [138] Zhou Y, Hao F, Meng W, Fu J. Scenario analysis of energy-based low-carbon development in China. J Environ Sci 2014;26(8):1631–40.
- [139] Zhou Y, Zhang B, Zou J, Bi J, Wang K. Joint R & D in low-carbon technology development in China: a case study of the wind-turbine manufacturing industry. Energy Policy 2012;46:100-8.
- [140] Zhu H, Deng Y, Zhu R, He X. Fear of nuclear power? Evidence from Fukushima nuclear accident and land markets in China. Reg Sci Urban Econ 2015:60:139-54.
- [141] Zhu J, Tang ZZ, Liu XM, Lu YF. Evaluation for Low Carbon City based on DPSIR Model: empirical research in Jiyuan City. Urban Probl 2012(12):42–7, [in Chinese].
- [142] Zhu L, Guan D, Moore S, Lee H, Su J, Zhang Q. Climate policy: steps to China's carbon peak. Nature 2015;522:279–81.
- [143] Zhu Q, Peng X, Wu K. Calculation and decomposition of indirect carbon emissions from residential consumption in china based on the input–output model. Energy Policy 2012;48(3):618–26.
- [144] Zhu Q, Lowe EA, Barnes D. Industrial symbiosis in China: a case study of the Guitang Group. J Ind Ecol 2007;11.1:31–42.
- [145] Zielke P, Waibel M. The governance of spatial-economic restructuring in contemporary China: the case of textile clusters. Pac Geogr 2014;42:21–7.
- [146] Zielke P, Waibel M. Urban governance of economic restructuring processes in China: the case of the Guangzhou Zhongda Textile district. ASIEN – Ger J Contemp Asia 2012;125:35–61.